Author’s response to reviews

Title: Hepatobiliary and Pancreatic Manifestations in Inflammatory Bowel Diseases: A Referral Center Study

Authors:

Fotios Fousekis (fotisfous@gmail.com)
Konstantinos Katsanos (khkostas@hotmail.com)
Vasileios Theopistos (vasileios.theopistos@gmail.com)
Gerasimos Baltayiannis (gmpaltag@cc.uoi.gr)
Maria Kosmidou (mkosmidou@uhi.gr)
Georgios Glantzounis (gglantzounis@gmail.com)
Leonidas Christou (lchristu@cc.uoi.gr)
Epameinondas Tsianos (ep.tsianos@gmail.com)
Dimitrios Christodoulou (dchristodoulou@gmail.com)

Version: 3 Date: 23 Mar 2019

Author’s response to reviews:

Ioannina, 23/03/2019

To BMC Gastroenterology

Dear Editor,

I hereby submit for your consideration the revised article, titled “Hepatobiliary and Pancreatic Manifestations in Inflammatory Bowel Diseases: A Referral Center Study”. My colleagues — co-authors and I sincerely hope that you will deem it appropriate for publication in BMC Gastroenterology.

We would like to thank the Reviewers for their useful comments.
Responses to reviewer and Editor

Comments:

1) Declarations. Please add a "Declarations" heading to your Declarations section. The Declarations should follow the Conclusions section, and be before the References.

2) Competing Interests. In order to be in line with journal requirements please change your “Conflict of Interest” heading to instead read “Competing Interests”.

3) Authors Contributions. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Please represent authors’ names using their initials, not their full name, in the Authors’ Contributions section of your Declarations.

4) Funding. We note that you have not included a ‘Funding’ section in the Declarations. All sources of funding for the research reported should be declared. If no funding was obtained for your study we still require this section to be included with the statement “No funding was obtained for this study”. In the “Funding” section in the Declarations, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

5) Clean Manuscript. Please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please also ensure that all figures, tables, and supplementary/additional files are cited in the text.

Response:

On comment 1: It was done.

On comment 2: It was done.

On comment 3: We added on Authors contributions, “ET, DC, KK, Research concept and design. FF, VT, KK, GB Collection of data. MK, GG, Data Analysis, FF, ET, DC, LC, Writing”.

On comment 4: It was done.

On comment 5: It was done.

We hope that the revised manuscript will vouch for a position among other relative studies already published in your Journal. Once more we would like to thank the Section Editor and their reviewers for their fruitful comments and their important contribution.
On behalf of the co-authors with my great appreciation

Dimitrios Christodoulou, MD, PhD, FEBGH
Professor of Gastroenterology